Author’s response to reviews

Title: Tardive Dyskinesia in an Adolescent Male Abusing Quetiapine via Insufflation: A Case Study

Authors:

- Mathew George (Drmg456@gmail.com)
- Maya Haasz (mayahaasz@gmail.com)
- Alvaro Coronado (alvaro40@yahoo.com)
- Steven Salhanick (ssalhani@bidmc.harvard.edu)
- Lindsey Korbel (lindsey.korbel@osumc.edu)
- Joseph P Kitzmiller (joseph.kitzmiller@osumc.edu)

Version: 2 Date: 29 April 2013

Author’s response to reviews: see over
April 29, 2013

Dear BMC Pediatrics Editor(s)

Please find attached our revised case report. The following requested changes have been addressed:

1) Please update the patient consent statement. The following statement is now in the consent section: “Written informed consent was obtained from the patient’s legal guardian and from the patient for publication of this case report. A copy of the written consent is available for review by the Editor of this journal.”

2) Please revise the Competing Interests Section. This section is now placed after the Conclusions/Abbreviations. The section reads as follows: “The authors declare that they do not have any financial or any non-financial (political, personal, religious, ideological, academic, intellectual, commercial or any other) competing interests regarding this manuscript.”

3) The Acknowledgement Section now states “The authors have no acknowledgments regarding this manuscript. There was no funding source for the writing of this manuscript.”

Regards,

Joseph Kitzmiller MD PhD FCP CPI
Research Assistant Professor
Departments of Pharmacology and Biomedical Engineering
The Ohio State University Medical Center
joseph.kitzmiller@osumc.edu
(614) 292 8438
(614) 481-3105